Author's response to reviews

Title: Molecular Genetics Analysis of Hereditary Breast and Ovarian Cancer Patients in India

Authors:

NAGASAMY SOUMITTRA (soumi_n@yahoo.co.uk)
BALAIAH MEENAKUMARI (hema_meenakumari@yahoo.co.in)
TITHI PARIJA (tithiparija@yahoo.com)
VELUSWAMI SRIDEVI (sri04devi@yahoo.co.in)
KARUNAKARAN N NANCY (nirc18@gmail.com)
RAJARAMAN SWAMINATHAN (rcsurvival@yahoo.co.uk)
KAMALALAYAM R RAJALEKSHMY (rajalekshmy_raghavan@yahoo.com)
URMILA MAJHI (umajhi@hotmail.com)
THANGARAJAN RAJKUMAR (drtrajkumar@gmail.com)

Version: 4 Date: 27 July 2009

Author's response to reviews:

27-07-09

Sir,

Many thanks for accepting our manuscript for publication. The following changes which had been requested have been made.

1. Author list will be the same in both the submission system and in the manuscript.
2. The figures in the manuscript files have been removed.
3. Figure 4 had been cited in the manuscript, page 8, the last para of the results section.
4. The references have been amended to include all authors.
5. The references have been cited in the text as required.

The name of the web site for Reference 19 has been included.

6. The vertical lines in Table will not be visible.

Thanking you,

Yours sincerely,

Dr.T.Rajkumar